UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported) April 19, 2021

BRICKELL BIOTECH, INC.

(Exact name of Registrant as specified in its charter)

Delaware

(State or Other Jurisdiction of Incorporation)

000-21088 (Commission File Number) 93-0948554 (IRS Employer Identification No.)

5777 Central Avenue Suite 102 Boulder, CO 80301 (Address of Principal Executive Offices) (Zip Code)

Registrant's telephone number, including area code: (720) 505-4755

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (*see* General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common stock, par value \$0.01 per share	BBI	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers

As reported below in Item 5.07, on April 19, 2021, the stockholders of Brickell Biotech, Inc. (the "Company") approved amendments to the Company's 2020 Omnibus Long-Term Incentive Plan (the "Omnibus Plan") to increase the number of shares of the Company's common stock, par value \$0.01 per share (the "Common Stock"), authorized for issuance under the Omnibus Plan, and the number of such shares that can be delivered in respect of incentive stock options, by 4,000,000 shares (the "Plan Amendments"). The Plan Amendments were included as Proposal 4 in the Company's definitive proxy statement for its Annual Meeting of Stockholders filed with the Securities and Exchange Commission on March 15, 2021 (the "Proxy Statement"). A copy of the Omnibus Plan, as amended to reflect the Plan Amendments, is attached as Exhibit 10.1 to this Current Report on Form 8-K and is incorporated herein by reference.

As reported below in Item 5.07, on April 19, 2021, the stockholders of the Company approved the Company's Employee Stock Purchase Plan (the "ESPP"). Approval of the ESPP was included as Proposal 5 in the Proxy Statement. A copy of the ESPP is attached as Exhibit 10.2 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year

As reported below in Item 5.07, on April 19, 2021, the stockholders of the Company approved an amendment to the Company's Restated Certificate of Incorporation (the "Certificate of Incorporation") to increase the number of authorized shares of Common Stock from 100,000,000 to 300,000,000 (the "Charter Amendment"). The Charter Amendment, which was included as Proposal 3 in the Proxy Statement, became effective on April 19, 2021 upon filing of a Certificate of Amendment to the Certificate of Incorporation with the Secretary of State of the State of Delaware.

A copy of the Certificate of Amendment as filed with the Secretary of State of the State of Delaware is attached as Exhibit 3.1 to this Current Report on Form 8-K and is incorporated herein by reference. A copy of the Company's Certificate of Incorporation, as amended and restated to reflect the Charter Amendment, is attached as Exhibit 3.2 to this Current Report on Form 8-K and is incorporated herein by reference.

Item 5.07 Submission of Matters to a Vote of Security Holders

On April 19, 2021, the Company held its 2021 Annual Meeting of Stockholders (the "Annual Meeting"). The following matters (the "Proposals") were voted upon by the stockholders with the final voting results as shown:

Proposal 1: To elect two (2) directors to serve as Class II directors until the 2024 Annual Meeting of Stockholders and until their respective successors are elected.

Name	For	Withhold	Broker Non-Votes
Robert B. Brown	15,843,494	3,297,564	21,050,783
Dennison T. Veru	17,592,541	1,548,517	21,050,783

Proposal 2 – To ratify the selection of Ernst & Young LLP to serve as the Company's registered independent public accounting firm for the year ending December 31, 2021.

For	Against	Abstain	Broker Non-Votes
39,528,838	436,952	226,051	_

Proposal 3 - To approve an amendment to the Certificate of Incorporation to increase the number of authorized shares of Common Stock from 100,000,000 to 300,000,000.

For	Against	Abstain	Broker Non-Votes
34,287,450	5,451,640	452,751	—

Proposal 4 – To approve amendments to the Omnibus Plan to increase the number of shares of Common Stock authorized for issuance under the Omnibus Plan, and the number of such shares that can be delivered in respect of incentive stock options, by 4,000,000 shares.

 For	Against	Abstain	Broker Non-Votes
13,552,065	5,281,163	307,830	21,050,783

Proposal 5 - To approve the ESPP.

For	Against	Abstain	Broker Non-Votes
17,318,439	1,424,065	398,554	21,050,783

Proposal 6 – To approve the issuance of more than 20% of the Common Stock pursuant to a purchase agreement with Lincoln Park Capital Fund, LLC, for purposes of Nasdaq Listing Rule 5635(d).

For	Against	Abstain	Broker Non-Votes
16,012,540	2,875,479	253,039	21,050,783

All six Proposals were approved, each receiving the affirmative requisite vote of the holders of shares of Common Stock.

Each Proposal is described in detail in the Proxy Statement.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

3.1 Certificate of Amendment to the Restated Certificate of Incorporation, dated April 19, 2021

3.2 Amended and Restated Certificate of Incorporation, as amended through April 19, 2021

10.1 Brickell Biotech, Inc. 2020 Omnibus Long-Term Incentive Plan, as amended through April 19, 2021

10.2 Brickell Biotech, Inc. Employee Stock Purchase Plan

104 Cover Page Interactive Data File (embedded within the Inline XBRL document)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

By:

Date: April 19, 2021

Brickell Biotech, Inc.

/s/ Robert B. Brown Name: Robert B. Brown Title: Chief Executive Officer

CERTIFICATE OF AMENDMENT TO THE CERTIFICATE OF INCORPORATION OF BRICKELL BIOTECH, INC.

Brickell Biotech, Inc., a corporation organized and existing under the General Corporation Law of the State of Delaware (the "<u>Corporation</u>"), hereby certifies as follows:

FIRST: At a meeting of the Board of Directors of the Corporation, the Board of Directors approved a proposed amendment to the Restated Certificate of Incorporation (the "<u>Certificate of Incorporation</u>") of the Corporation and directed that such amendment be submitted to the Corporation's stockholders for their consideration at the Corporation's 2021 annual meeting of stockholders, with a recommendation from the Board of Directors that the stockholders vote for approval of such amendment.

SECOND: The proposed amendment provides that Article IV, Section A of the Certificate of Incorporation is hereby amended to read in its entirety as follows:

ARTICLE IV

A. <u>Classes of Stock</u>. The total number of shares of all classes of capital stock which the corporation shall have authority to issue is Three Hundred Five Million (305,000,000) shares, of which Three Hundred Million (300,000,000) shares of the par value of One Cent (\$0.01) each shall be Common Stock (the "Common Stock") and Five Million (5,000,000) shares of the par value of One Cent (\$0.01) each shall be Preferred Stock (the "Preferred Stock").

The Preferred Stock may be issued from time to time in one or more series. The Board of Directors is authorized to fix the number of shares of any series of Preferred Stock and to determine the designation of any such shares. The Board of Directors also is authorized to determine or alter the rights (including but not limited to voting rights), preferences, privileges and restrictions granted to or imposed upon any wholly unissued series of Preferred Stock, and within the limits and restrictions stated in any resolution or resolutions of the Board of Directors originally fixing the number of shares constituting any series, to increase or decrease (but not below the number of shares of such series subsequent to the issue of shares of that series by filing a certificate pursuant to the applicable laws of the State of Delaware.

THIRD: Pursuant to Section 242 of the General Corporation Law of the State of Delaware, at the Corporation's 2021 annual meeting of stockholders, duly called and held upon notice in accordance with Section 222 of the General Corporation Law of the State of Delaware, the necessary number of shares as required by statute were voted in favor of the amendment.

FOURTH: The foregoing amendment was duly adopted in accordance with Section 242 of the General Corporation Law of the State of Delaware.

FIFTH: The foregoing amendment shall be effective on the date this Certificate of Amendment is filed with the Secretary of State of the State of Delaware.

[Signature on Following Page]

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Amendment to be signed by the authorized officer named below, this 19th day of April, 2021.

By: <u>/s/ Robert B. Brown</u> Name: Robert B. Brown Title: Chief Executive Officer

[Signature Page to Certificate of Amendment]

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION, OF BRICKELL BIOTECH, INC.

(Composite document reflecting amendments through April 19, 2021)

ARTICLE I

The name of the corporation is Brickell Biotech, Inc.

ARTICLE II

The address of its registered office in the State of Delaware is Corporation Trust Center, 1209 Orange Street, in the City of Wilmington, County of New Castle, Delaware 19801. The name of its registered agent at such address is The Corporation Trust Company.

ARTICLE III

The corporation is organized for the purposes of transacting any or all lawful business for which corporations may be organized under the laws of the United States and the laws of the State of Delaware. The corporation shall have all of the corporate powers enumerated under Delaware law.

ARTICLE IV

A. <u>Classes of Stock</u>. The total number of shares of all classes of capital stock which the corporation shall have authority to issue is Three Hundred Five Million (305,000,000) shares, of which Three Hundred Million (300,000,000) shares of the par value of One Cent (\$0.01) each shall be Common Stock (the "Common Stock") and Five Million (5,000,000) shares of the par value of One Cent (\$0.01) each shall be Preferred Stock (the "Preferred Stock").

The Preferred Stock may be issued from time to time in one or more series. The Board of Directors is authorized to fix the number of shares of any series of Preferred Stock and to determine the designation of any such shares. The Board of Directors also is authorized to determine or alter the rights (including but not limited to voting rights), preferences, privileges and restrictions granted to or imposed upon any wholly unissued series of Preferred Stock, and within the limits and restrictions stated in any resolution or resolutions of the Board of Directors originally fixing the number of shares constituting any series, to increase or decrease (but not below the number of shares of such series outstanding) the number of shares of such series subsequent to the issue of shares of that series by filing a certificate pursuant to the applicable laws of the State of Delaware.

B. Common Stock.

1. <u>Relative Rights of Preferred Stock and Common Stock</u>. All preferences, voting powers, and relative, participating, optional or other special rights and privileges, and qualifications, limitations, or restrictions of the Common Stock are made expressly subject and subordinate to those that may be fixed with respect to any shares of the Preferred Stock.

2. <u>Voting Rights</u>. Except as otherwise required by law or this Restated Certificate of Incorporation, each holder of Common Stock shall have one vote in respect of each share of stock held by such holder of record on the books of the corporation for the election of directors and on all matters submitted by the Board of Directors to a vote of stockholders of the corporation.

3. <u>Dividends</u>. Subject to the preferential rights of the Preferred Stock, the holders of shares of Common Stock shall be entitled to receive, when and if declared by the Board of Directors, out of the assets of the corporation which are by law available therefor, dividends payable either in cash, in property or in shares of capital stock.

4. <u>Dissolution, Liquidation or Winding Up</u>. In the event of any dissolution, liquidation or winding up of the affairs of the corporation, after distribution in full of the preferential amounts, if any, to be distributed to the holders of shares of the Preferred Stock, holders of Common Stock shall be entitled, unless otherwise provided by law or this Restated Certificate of Incorporation, to receive all of the remaining assets of the corporation of whatever kind available for distribution to stockholders ratably in proportion to the number of shares of Common Stock held by each of them respectively.

ARTICLE V

A. <u>Board Size</u>. The number of directors that constitutes the entire Board shall be fixed by, or in the manner provided in, the Amended and Restated Bylaws of the Corporation. At each annual meeting of stockholders, directors of the Corporation shall be elected to hold office until the expiration of the term for which they are elected and until their successors have been duly elected and qualified or until their earlier death, resignation or removal; except that if any such election shall not be so held, such election shall take place at a stockholders' meeting called and held in accordance with the Delaware General Corporation Law

B. <u>Board Structure</u>. From and after the Effective Time, the directors shall be divided into three (3) classes as nearly equal in size as is practicable, hereby designated Class I, Class II and Class III. The Board may assign members of the Board already in office to such classes at the time such classification becomes effective. The term of office of the initial Class III directors shall expire at the annual meeting of the stockholders in 2019, the term of office of the initial Class I directors shall expire at the annual meeting of the stockholders in 2020, and the term of

office of the initial Class II directors shall expire at the annual meeting of the stockholders in 2021. At each annual meeting of stockholders, commencing with the first regularly scheduled annual meeting of stockholders in 2019, each of the successors elected to replace the directors of a Class whose term shall have expired at such annual meeting shall be elected to hold office for a three year term and until the third annual meeting next succeeding his or her election and until his or her respective successor shall have been duly elected and qualified. Notwithstanding the foregoing provisions of this Article V, each director shall serve until his or her successor is duly elected and qualified or until his or her death, resignation, or removal. If the number of directors is thereafter changed, any newly created directorships or decrease in directorships shall be so apportioned among the classes as to make all classes as nearly equal in number as is practicable. No decrease in the number of directors constituting the Board shall shorten the term of any incumbent director.

C. <u>Removal</u>; <u>Vacancies</u>. Any director may be removed from office by the stockholders of the Corporation as provided in Section 141(k) of the Delaware General Corporation Law. Vacancies occurring on the Board for any reason and newly created directorships resulting from an increase in the authorized number of directors may be filled only by vote of a majority of the remaining members of the Board, although less than a quorum, or by a sole remaining director, and not by stockholders. A person elected to fill a vacancy or newly created directorship shall hold office until the next election of the class for which such director shall have been chosen and until his or her successor shall be duly elected and qualified.

ARTICLE VI

Special meetings of the stockholders of the corporation, unless otherwise prescribed by applicable law or by this certificate, may be called only by the Chief Executive Officer of the corporation or by a resolution adopted by the affirmative vote of a majority of a quorum of the Board of Directors.

ARTICLE VII

Election of directors need not be by written ballot unless the Bylaws so provide.

ARTICLE VIII

The Board of Directors is empowered expressly to adopt, amend or repeal the Bylaws of the corporation; provided, however, that any adoption, amendment or repeal of the Bylaws of the corporation by the Board of Directors shall require the approval of at least sixty-six and two-thirds percent (66 2/3%) of the total number of authorized directors (whether or not there exist any vacancies in previously authorized directorships at the time any resolution providing for adoption, amendment or repeal is presented to the Board of Directors). The stockholders also shall have power to adopt, amend or repeal the Bylaws of the corporation; provided, however, that in addition to any vote of the holders of any class or series of stock of the corporation

required by law or by this Restated Certificate of Incorporation the affirmative vote of the holders of at least sixty-six and two-thirds percent $(66\ 2/3\%)$ of the voting power of all of the then outstanding shares of the stock of the corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for such adoption, amendment or repeal by the stockholders of any provision(s) of the Bylaws of the corporation.

ARTICLE IX

One third (1/3) of the shares entitled to be cast on any matter by a voting group, or in case of the Board, the number of directors shall constitute a quorum of that voting group, or the Board, for action on that matter.

ARTICLE X

No contract or other transaction between the corporation and one or more of its directors, or between the corporation and any other corporation, firm, association or other entity in which one or more of the directors are directors or officers, or are financially interested, shall be either void or voidable because of such relationship or interest or because such director or directors are present at the meeting of the Board of Directors or a committee thereof which authorizes, approves or ratifies such contract or transaction or because his or her votes are counted for such purpose, if:

A. The material facts of such relationship or interest are disclosed or known to the Board of Directors, or a duly empowered committee thereof, which in good faith authorizes, approves or ratifies the contract or transaction by a majority vote or written consent sufficient for such purpose without counting the vote or votes of such interested director or directors, even though the disinterested directors comprise less than a quorum; or

B. The material facts of such relationship or interest are disclosed or known to the shareholders entitled to vote and they in good faith by majority vote of a quorum of the shareholders or written consent authorize, approve or ratify such contract or transaction; or

C. The contract or transaction is fair and reasonable as to the corporation at the time it is authorized by the Board of Directors, a committee thereof, or the shareholders.

A director of the corporation may transact business, borrow, lend, or otherwise deal or contract with the corporation to the full extent and subject only to the limitations and provisions of the applicable laws of the State of Delaware and the laws of the United States.

Interested directors may be counted in determining the presence of a quorum at a meeting of the Board of Directors or a committee thereof which authorizes, approves or ratifies such contract or transaction.

ARTICLE XI

A. <u>No Personal Liability</u>. The liability of a director of the corporation for monetary damages shall be eliminated to the fullest extent under applicable law.

B. <u>Indemnification</u>. To the fullest extent permitted by applicable law, the corporation is authorized to provide indemnification of (and advancement of expenses to) directors, officers and agents of the corporation (and any other persons to which applicable law permits the corporation to provide indemnification) through Bylaw provisions, agreements with such agents or other persons, vote of stockholders or disinterested directors or otherwise in excess of the indemnification and advancement otherwise permitted by such applicable law. If applicable law is amended after approval by the stockholders of this Article XI to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director to the corporation shall be eliminated or limited to the fullest extent permitted by applicable law as so amended.

Any repeal or modification of this Article shall only be prospective and shall not affect the rights or protections or increase the liability of any director under this Article in effect at the time of the alleged occurrence of any act or omission to act giving rise to liability or indemnification.

ARTICLE XII

Notwithstanding any other provision of this Restated Certificate of Incorporation, the affirmative vote of the holders of at least sixty-six and two-thirds percent (66-2/3%) of the voting power of all of the then outstanding shares of the stock of the corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required to amend in any respect or repeal Article XI, or Articles V, VI, VIII and X, hereof.

Exhibit 10.1

BRICKELL BIOTECH, INC. 2020 OMNIBUS LONG-TERM INCENTIVE PLAN, AS AMENDED ON APRIL 19, 2021

TABLE OF CONTENTS

		Page
SECTION 1	GENERAL	1
1.1.	Purpose	1
1.2.	Participation	1
1.3.	Foreign Participants	1
1.4.	Operation and Administration	1
1.5.	History	2
SECTION 2	DEFINITIONS	2
SECTION 3	SHARES AND PLAN LIMITS	6
3.1.	Shares of Stock and Other Amounts Subject to Plan	6
3.2.	Adjustments	8
3.3.	Plan Limitations	9
SECTION 4	OPTIONS	9
4.1.	Grant of Options	9
4.2.	Option Agreement	10
4.3.	Term of Option	10
4.4.	Exercise Price	10
4.5.	Minimum Vesting	10
4.6.	Payment of Option Exercise Price	10
4.7.	No Repricing	11
SECTION 5	FULL VALUE AWARDS	11
5.1.	Grant of Full Value Award	11
5.2.	Full Value Award Agreement	12
5.3.	Conditions	12
5.4.	Minimum Vesting	12
SECTION 6	CASH INCENTIVE AWARDS	13
SECTION 7	CHANGE IN CONTROL	13
7.1.	Change in Control	13
7.2.	Committee Actions On a Change in Control	13
SECTION 8	COMMITTEE	14
8.1.	Administration	14
8.2.	Selection of Committee	14
8.3.	Powers of Committee	14
8.4.	Delegation by Committee	15
8.5.	Information to be Furnished to Committee	15
8.6.	Liability and Indemnification of Committee	16
SECTION 9	AMENDMENT AND TERMINATION	16
SECTION 10	GENERAL PROVISIONS	17

ii

10.1.	General Restrictions	17
10.2.	Tax Withholding	17
10.3.	Grant and Use of Awards	18
10.4.	Dividends and Dividend Equivalents	18
10.5.	Settlement of Awards	19
10.6.	Transferability	19
10.7.	Form and Time of Elections	19
10.8.	Agreement With Company	19
10.9.	Action by Company or Subsidiary	20
10.10.	Gender and Number	20
10.11.	Limitation of Implied Rights	20
10.12.	Evidence	21
10.13.	Limitations under Section 409A	21

iii

BRICKELL BIOTECH, INC. 2020 OMNIBUS LONG-TERM INCENTIVE PLAN

SECTION 1 GENERAL

1.1. <u>Purpose</u>. The Brickell Biotech, Inc. 2020 Omnibus Long-Term Incentive Plan (the "Plan") has been established by Brickell Biotech, Inc., a Delaware corporation, (the "Company") to (i) attract and retain persons eligible to participate in the Plan; (ii) motivate Participants, by means of appropriate incentives, to achieve long-range goals; (iii) provide incentive compensation opportunities that are competitive with those of other similar companies; and (iv) further align the interests of Participants with those of the Company's other stockholders through compensation that is based on the Company's shares; and thereby promote the long-term financial interest of the Company and the Related Companies including the growth in value of the Company's shares and enhancement of long-term stockholder return. Capitalized terms in the Plan are defined in Section 2.

1.2. <u>Participation</u>. Subject to the terms and conditions of the Plan, the Committee shall determine and designate, from time to time, from among the Eligible Individuals, those persons who will be granted one or more Awards under the Plan, and thereby become "Participants" in the Plan.

1.3. Foreign Participants. In order to assure the viability of Awards granted to Participants who are subject to taxation in foreign countries, the Committee may provide for such special terms as it may consider necessary or appropriate to accommodate differences in local law, tax policy, or custom. Moreover, the Committee may approve such appendixes, supplements to, or amendments, restatements, or alternative versions of, the Plan as it may consider necessary or appropriate for such purposes without thereby affecting the terms of the Plan as in effect for any other purpose; provided, however, that no such supplements, amendments, restatements, or alternative versions shall increase the share limitations contained in Section 3.1 of the Plan.

1.4. <u>Operation and Administration</u>. The operation and administration of the Plan, including the Awards made under the Plan, shall be subject to the provisions of Section 8 (relating to operation and administration).

1.5. <u>History</u>. The Plan was adopted by the Company on March 16, 2020, subject to approval by stockholders. To the extent not prohibited by Applicable Laws, Awards which are to use shares of Stock reserved under the Plan that are contingent on the approval by the Company's stockholders may be granted prior to that meeting contingent on such approval. The Plan shall be unlimited in duration and, in the event of Plan termination, shall remain in effect as long as any Awards under it are outstanding; provided, however, that no Awards may be granted under the Plan after the ten-year anniversary of the date on which the stockholders approved the Plan.

SECTION 2

DEFINITIONS

2.1. "Administrator" means the Board or any of its Committees as will be administering the Plan, in accordance with Section 8.

2.2. "Applicable Laws" means the requirements relating to the administration of equity-based awards under U.S. state corporate laws, U.S. federal and state securities laws, the Code, any stock exchange or quotation system on which the Common Stock is listed or quoted and the applicable laws of any foreign country or jurisdiction where Awards are, or will be, granted under the Plan.

2.3. "Award Agreement" means the written agreement, including an electronic agreement, setting forth the terms and conditions applicable to each Award granted under the Plan. The Award Agreement is subject to the terms and conditions of the Plan.

2.4. "Award" means any award or benefit granted under the Plan, including, without limitation, the grant of Options and Full Value Awards.

2.5. "Board" means the Board of Directors of the Company.

- 2.6. "Change in Control" means the first to occur of any of the following:
- (a) the consummation of a purchase or other acquisition by any person, entity or group of persons (within the meaning of Section 13(d) or 14(d) of the Exchange Act or any comparable successor provisions, other than an acquisition by a trustee or other fiduciary holding securities under an employee benefit plan or similar plan of the Company or a Related Company), of "beneficial ownership" (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 50% or more of either the

outstanding shares of Stock or the combined voting power of the Company's then outstanding voting securities entitled to vote generally;

- (b) the consummation of a reorganization, merger, consolidation, acquisition, share exchange or other corporate transaction of the Company, in each case, with respect to which persons who were stockholders of the Company immediately prior to such reorganization, merger or consolidation do not, immediately thereafter, own more than 50% of the combined voting power entitled to vote generally in the election of directors of the reorganized, merged or consolidated company's then outstanding securities;
- (c) the consummation of any plan of liquidation or dissolution of the Company providing for the sale or distribution of substantially all of the assets of the Company and its Subsidiaries or the consummation of a sale of substantially all of the assets of the Company and its Subsidiaries; or
- (d) at any time during any period of two consecutive years, individuals who at the beginning of such period were members of the Board cease for any reason to constitute at least a majority thereof (unless the election, or the nomination for election by the Company's stockholders, of each new director was approved by a vote of at least two-thirds of the directors still in office at the time of such election or nomination who were directors at the beginning of such period).

2.7. "Code" means the United States Internal Revenue Code of 1986, as amended. A reference to any provision of the Code shall include reference to any successor provision of the Code.

- 2.8. "Committee" has the meaning set forth in Section 8.1.
- 2.9. "Common Stock" or "Stock" means the common stock of the Company.
- 2.10. "Company" has the meaning set forth in Section 1.1.

2.11. "Consultant" means any natural person engaged as a consultant or advisor by the Company or a Parent or Subsidiary or other Related Company (as determined by the Committee) to render bona fide services to such entity and such services are not in connection with the sale of shares of Stock in a capital-raising transaction, and do not directly or indirectly promote or maintain a market for the Company's securities. 2.12. "Director" means a member of the Board.

2.13. "Eligible Individual" means any Employee, Consultant or Director; provided, however, that to the extent required by the Code, an ISO may only be granted to an Employee of the Company or a Parent or Subsidiary. An Award may be granted to an Employee, Consultant or Director, in connection with hiring, retention or otherwise, prior to the date the Employee, Consultant or Director first performs services for the Company or the Subsidiaries, provided that such Awards shall not become vested prior to the date the Employee, Consultant or Director first performs such services.

2.14. "Employee" means any person, including officers and Directors, employed by the Company or any Parent or Subsidiary of the Company or a Related Company (as determined by the Committee). Neither service as a Director nor payment of a director's fee by the Company will be sufficient to constitute "employment" by the Company.

2.15. "Exchange Act" means the Securities Exchange Act of 1934, as amended.

2.16. "Exercise Price" of each Option granted under this Plan shall be established by the Committee or shall be determined by a method established by the Committee at the time the Option is granted.

2.17. "Fair Market Value" means, as of any date, the value of Common Stock determined as follows:

- (a) If the Common Stock is listed on any established stock exchange or a national market system, including without limitation the Nasdaq Stock Exchange, its Fair Market Value will be the closing sales price for such stock (or the closing bid, if no sales were reported) as quoted on such exchange or system on the last previous trading day prior to such date of determination, as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable;
- (b) If the Common Stock is regularly quoted by a recognized securities dealer but selling prices are not reported, the Fair Market Value of a share of Stock will be the mean between the high bid and low asked prices for the Common Stock on the last previous trading day prior to such date of determination (or, if no bids and asks were reported on that date, as applicable, on the last trading date such bids and asks were reported), as reported in *The Wall Street Journal* or such other source as the Administrator deems reliable; or

(c) In the absence of an established market for the Common Stock, the Fair Market Value will be determined in good faith by the Administrator.

2.18. A "Full Value Award" is a grant of one or more shares of Stock or a right to receive one or more shares of Stock in the future, with such grant subject to one or more conditions, as determined by the Committee.

2.19. An "Incentive Stock Option" or an "ISO" is an Option that is intended to satisfy the requirements applicable to an "incentive stock option" described in Section 422(b) of the Code.

2.20. A "Non-Qualified Option or an "NQO" is an Option that is not intended to be an "incentive stock option" as that term is described in Section 422(b) of the Code.

2.21. An "Option" entitles the Participant to purchase shares of Stock at an Exercise Price established by the Committee. Any Option granted under this Plan may be either an ISO or an NQO as determined in the discretion of the Committee.

2.22. "Outside Director" means a Director of the Company who is not an officer or employee of the Company or the Related Companies.

2.23. "Parent" means a parent corporation within the meaning of Section 424(e) of the Code.

2.24. "Participant" means the holder of an outstanding Award.

2.25. "Period of Restriction" means the period during which the transfer of shares of Stock are subject to restrictions and therefore, the shares of Stock are subject to a substantial risk of forfeiture. Such restrictions may be based on the passage of time, the achievement of target levels of performance, or the occurrence of other events as determined by the Administrator.

2.26. "Plan" has the meaning set forth in Section 1.1.

2.27. "Related Company" means any corporation, partnership, joint venture, limited liability company or other entity during any period in which a controlling interest in such entity is owned, directly or indirectly, by the Company (or by any entity that is a successor to the Company), and any other business venture designated by the Committee in which the Company (or any entity that is a successor to the

Company) has, directly or indirectly, a significant interest (whether through the ownership of securities or otherwise), as determined in the discretion of the Committee.

2.28. "Securities Act" means the Securities Act of 1933, as amended.

2.29. "Subsidiary" means a "subsidiary corporation," whether now or hereafter existing, as defined in Section 424(f) of the Code.

2.30. "Termination Date" means the date on which a Participant both ceases to be an employee of the Company and the Related Companies and ceases to perform material services for the Company and the Related Companies (whether as a director or otherwise), regardless of the reason for the cessation; provided that a "Termination Date" shall not be considered to have occurred during the period in which the reason for the cessation of services is a leave of absence approved by the Company or the Related Company which was the recipient of the Participant's services; and provided, further that, with respect to an Outside Director, "Termination Date" means the date on which the Outside Director's service as an Outside Director terminates for any reason. If, as a result of a sale or other transaction, the entity for which the Participant performs services ceases to be a Related Company (and such entity is or becomes an entity separate from the Company), the occurrence of such transaction shall be the Participant's termination Date. With respect to Awards that constitute deferred compensation subject to Section 409A of the Code, references to the Participant's separation from service, and other similar reference) and references to a Participant's termination as a Director (including separation from service and other similar references) shall mean the date that the Participant incurs a "separation from service" within the meaning of Section 409A of the Code.

SECTION 3 SHARES OF STOCK AND PLAN LIMITS

3.1. <u>Shares of Stock and Other Amounts Subject to Plan</u>. The shares of Stock for which Awards may be granted under the Plan shall be subject to the following:

(a) Subject to the following provisions of this Section 3.1, the maximum number of shares of Stock that may be delivered to Participants and their beneficiaries under the Plan shall be the sum of (i) 9,179,389 shares of Stock (which number includes all shares available for delivery under this Section 3.1(a) since the establishment of the Plan, determined in accordance with the terms of the Plan); and (ii) any shares granted previously under the Company's 2009 Equity Incentive Plan, as amended and the Amended and Restated Stock Incentive Plan of Vical Incorporated (the "Prior Plans") that are forfeited, expire or are canceled after the Effective Date without delivery of shares or which result in the forfeiture of the shares back to the Company to the extent that such shares would have been added back to the reserve under the terms of the Prior Plans, but not including shares that remained available for grant pursuant to the Prior Plans that were not previously granted. Shares of Stock issued by the Company in connection with awards that are assumed or substituted in connection with a reorganization, merger, consolidation, acquisition, share exchange or other corporate transaction shall not be counted against the number of shares of Stock that may be issued with respect to Awards under the Plan.

- (b) Only shares of Stock, if any, actually delivered to the Participant or beneficiary on an unrestricted basis with respect to an Award shall be treated as delivered for purposes of the determination under Section 3.1(a) above, regardless of whether the Award is denominated in shares of Stock or cash. Consistent with the foregoing:
 - (i) To the extent any shares of Stock covered by an Award are not delivered to a Participant or beneficiary because the Award is forfeited or cancelled, or the shares of Stock are not delivered on an unrestricted basis (including, without limitation, by reason of the Award being settled in cash), such shares of Stock shall not be deemed to have been delivered for purposes of the determination under Section 3.1(a) above.
 - (ii) Subject to the provisions of paragraph (i) above, the total number of shares of Stock covered by an Award will be treated as delivered for purposes of this paragraph (b) to the extent payments or benefits are delivered to the Participant with respect to such shares. Accordingly (A) if shares covered by an Award are used to satisfy the applicable tax withholding obligation or Exercise Price, the number of shares held back by the Company to satisfy such withholding obligation or Exercise Price shall be considered to have been delivered; (B) if the Exercise Price of any Option granted under the Plan is satisfied by tendering shares of Stock to the Company (by either actual delivery or by attestation, including shares of Stock that would otherwise be distributable upon the exercise of the Option), the number of shares tendered



to satisfy such Exercise Price shall be considered to have been delivered; and (C) if shares of Stock are repurchased by the Company with proceeds received from the exercise of an option issued under this Plan, the total number of such shares repurchased shall be deemed delivered.

(c) The shares of Stock with respect to which Awards may be made under the Plan shall be: (i) shares currently authorized but unissued; (ii) to the extent permitted by Applicable Law, shares currently held or acquired by the Company as treasury shares, including shares purchased in the open market or in private transactions; or (iii) shares purchased in the open market by a direct or indirect wholly-owned subsidiary of the Company (as determined by the Chief Executive Officer or the Chief Financial Officer of the Company). The Company may contribute to the subsidiary or trust an amount sufficient to accomplish the purchase in the open market of the shares of Stock to be so acquired (as determined by the Chief Executive Officer or the Chief Financial Officer of the Company).

3.2. Adjustments. In the event of a corporate transaction involving the Company (including, without limitation, any share dividend, share split, extraordinary cash dividend, recapitalization, reorganization, merger, amalgamation, consolidation, share exchange, split-up, spin-off, sale of assets or subsidiaries, combination or exchange of shares), the Committee shall, in the manner it determines equitable in its sole discretion, adjust Awards to reflect the transactions. Action by the Committee may include: (i) adjustment of the number and kind of shares which may be delivered under the Plan; (ii) adjustment of the number and kind of shares subject to outstanding Awards; (iii) adjustment of the Exercise Price of outstanding Options; and (iv) any other adjustments that the Committee determines have comparable (which may include, without limitation, (A) replacement of Awards with other Awards which the Committee determines have comparable value and which are based on shares of a company resulting from the transaction, and (B) cancellation of the Award in return for cash payment of the current value of the Award, determined as though the Award is fully vested at the time of payment, provided that in the case of an Option, the amount of such payment will be the excess of value of the shares of Stock subject to the Option at the time of the transaction over the Exercise Price). However, in no event shall this Section 3.2 be construed to permit a modification (including a replacement) of an Option if such modification either: (i) would result in accelerated recognition of income or imposition of additional tax under Section 409A of the Code; or (ii) would cause the Option subject to the modification (or cause a replacement Option) to be subject to Section 409A of the Code, provided that the restriction of this clause (ii) shall not apply to any Option that, at the

time it is granted or otherwise, is designated as being deferred compensation subject to Section 409A of the Code.

- 3.3. <u>Plan Limitations</u>. Subject to Section 3.2, the following additional maximums are imposed under the Plan:
- (a) The maximum number of shares of Stock that may be delivered to Participants and their beneficiaries with respect to ISOs granted under the Plan shall be 9,179,389 shares of Stock (which number includes all shares of Stock available for delivery under this Section 3.3(a) since the establishment of the Plan, determined in accordance with the terms of the Plan).
- (b) Notwithstanding the provisions of Sections 4.5 and 5.4 of the Plan, the Committee may grant Awards that are not subject to the minimum vesting limitations of Sections 4.5 (with respect to Options) and of Section 5.4 (with respect to Full Value Awards) in certain circumstances as determined by the Committee in its sole discretion; provided, however, that the aggregate number of shares of Stock subject to Options and Full Value Awards granted pursuant to the Plan that are not subject to the minimum vesting limitations of Sections 4.5 and 5.4 (excluding any such Awards to the extent that they have been forfeited or cancelled) may not exceed 5% of the limit imposed by subsection 3.1(a) (relating to the limit on shares of Stock granted under the Plan).

SECTION 4

<u>OPTIONS</u>

4.1. <u>Grant of Options</u>. Subject to the terms and conditions of the Plan, the Administrator, at any time and from time to time, may grant Options to an Eligible Individual in such amounts as the Administrator, in its sole discretion, will determine. Each Option will be designated in the Award Agreement as either an ISO or an NQO. Notwithstanding a designation for a grant of Options as ISOs, however, to the extent that the aggregate Fair Market Value of the shares of Stock with respect to which ISOs are exercisable for the first time by the Participant during any calendar year (under all plans of the Company and any Parent or Subsidiary) exceeds \$100,000, such Options will be treated as NQOs. For purposes of this Section 4.1, ISOs will be taken into account in the order in which they were granted, the Fair Market Value of the shares of Stock with respect to such shares of Stock is granted, and calculation will be performed in accordance with Section 422 of the Code and Treasury Regulations promulgated thereunder.

4.2. <u>Option Agreement</u>. Each Award of an Option will be evidenced by an Award Agreement that will specify the date of grant of the Option, the Exercise Price, the term of the Option, the number of shares of Stock subject to the Option, the exercise restrictions, if any, applicable to the Option, including the dates upon which the Option is first exercisable in whole and/or part, and such other terms and conditions as the Administrator, in its sole discretion, may determine.

4.3. <u>Term of Option</u>. The term of each Option will be stated in the Award Agreement; provided, however, that the term will be no more than 10 years from the date of grant thereof. In the case of an ISO granted to a Participant who, at the time the ISO is granted, owns capital stock representing more than 10% of the total combined voting power of all classes of capital stock of the Company or any Parent or Subsidiary, the term of the ISO will be five years from the date of grant or such shorter term as may be provided in the Award Agreement.

4.4. <u>Exercise Price</u>. The Exercise Price shall not be less than 100% of the Fair Market Value of a share of Stock on the date of grant (or, if greater, the par value, if any, of a share of Stock). In addition, in the case of an ISO granted to an Employee who owns capital stock representing more than 10% of the voting power of all classes of capital stock of the Company or any Parent or Subsidiary, the per share Exercise Price will be no less than 110% of the Fair Market Value per share of Stock on the date of grant. Notwithstanding the foregoing provisions of this Section 4.4, Options may be granted with a per share Exercise Price of less than 100% of the Fair Market Value per share of Stock on the date of grant pursuant to a transaction described in, and in a manner consistent with, Section 424(a) of the Code.

4.5. <u>Minimum Vesting</u>. Notwithstanding the foregoing, and subject to Section 3.3(b), in no event shall an Option granted to any Participant become exercisable or vested prior to the first anniversary of the date on which it is granted (subject to acceleration of exercisability and vesting, to the extent permitted by the Committee, in the event of the Participant's death, disability, Change in Control or involuntary termination).

4.6. <u>Payment of Option Exercise Price</u>. The payment of the Exercise Price of an Option granted under this Section 4 shall be subject to the following:

(a) Subject to the following provisions of this Section 4.6, the full Exercise Price for shares of Stock purchased upon the exercise of any Option shall be paid at the time of such exercise (except that, in the case of an exercise arrangement approved by

the Committee and described in Section 4.6(c), payment may be made as soon as practicable after the exercise).

- (b) Subject to Applicable Law, the full Exercise Price shall be payable in cash, by promissory note, or by tendering, by either actual delivery of shares or by attestation, shares of Stock acceptable to the Committee (including shares otherwise distributable pursuant to the exercise of the Option), and valued at Fair Market Value as of the day of exercise, or in any combination thereof, as determined by the Committee.
- (c) Subject to Applicable Law, if shares are publicly traded, the Committee may permit a Participant to elect to pay the Exercise Price upon the exercise of an Option by irrevocably authorizing a third party to sell shares of Stock (or a sufficient portion of the shares of Stock) acquired upon exercise of the Option and remit to the Company a sufficient portion of the sale proceeds to pay the entire Exercise Price and any tax withholding resulting from such exercise.

4.7. No Repricing. Except for either adjustments pursuant to Section 3.2 (relating to the adjustment of shares of Stock), or reductions of the Exercise Price approved by the Company's stockholders, the Exercise Price for any outstanding Option may not be decreased after the date of grant nor may an outstanding Option granted under the Plan be surrendered to the Company as consideration for the grant of a replacement Option with a lower Exercise Price. Except as approved by Company's stockholders, in no event shall any Option granted under the Plan be surrendered to Company in consideration for a cash payment or the grant of any other Award if, at the time of such surrender, the Exercise Price of the Option is greater than the then current Fair Market Value of a share of Stock. In addition, no repricing of an Option shall be permitted without the approval of Company's stockholders if such approval is required under the rules of any stock exchange on which Stock is listed.

SECTION 5 FULL VALUE AWARDS

5.1. <u>Grant of Full Value Award</u>. Subject to the terms and conditions of the Plan, the Administrator, at any time and from time to time, may grant Full Value Awards to Eligible Individuals in such amounts as the Administrator, in its sole discretion, will determine.

5.2. <u>Full Value Award Agreement</u>. Each Full Value Award will be evidenced by an Award Agreement that will specify the Period of Restriction, the number of shares of Stock granted, and such other terms and conditions as the Administrator, in its sole discretion, may determine.

5.3. <u>Conditions</u>. A Full Value Award may be subject to one or more of the following, as determined by the Committee:

- (a) The grant shall be in consideration of a Participant's previously performed services, or surrender of other compensation that may be due.
- (b) The grant shall be contingent on the achievement of performance or other objectives during a specified period.
- (c) The grant shall be subject to a risk of forfeiture or other restrictions that will lapse upon the achievement of one or more goals relating to completion of service by the Participant, or achievement of performance or other objectives.

The grant of Full Value Awards may also be subject to such other conditions, restrictions and contingencies, as determined by the Committee.

5.4. Minimum Vesting.

- (a) Notwithstanding the foregoing, and subject to Section 3.3(b), if a Participant's right to become vested in a Full Value Award is conditioned on the completion of a specified period of service with the Company or the Related Companies, without achievement of performance targets or other performance objectives (whether or not related to performance measures) being required as a condition of vesting, and without it being granted in lieu of other compensation, then the required period of service for vesting shall be not less than one year (subject, to the extent provided by the Committee, to acceleration of vesting in the event of the Participant's death, disability, Change in Control or involuntary termination). The foregoing requirements shall not apply to grants that are a form of payment of earned performance awards or other incentive compensation.
- (b) Notwithstanding the foregoing, and subject to Section 3.3(b), if a Participant's right to become vested in a Full Value Award is conditioned on the achievement of performance targets or other performance objectives (whether or not related to performance measures and whether or not such Full Value Award is designated as "Performance-Based Compensation"), then the required performance period for

determining the achievement of such performance targets or other performance objectives for vesting shall be not less than one year (subject, to the extent provided by the Committee, to acceleration of vesting in the event of the Participant's death, disability, Change in Control or involuntary termination).

SECTION 6

CASH INCENTIVE AWARDS

6.1. A Cash Incentive Award is the grant of a right to receive a payment of cash (or in the discretion of the Committee, Shares having value equivalent to the cash otherwise payable) that is contingent on achievement of performance or other objectives over a specified period established by the Committee. The grant of Cash Incentive Awards may also be subject to such other conditions, restrictions and contingencies, as determined by the Committee. Except as otherwise provided in the applicable plan or arrangement, distribution of any bonus awards by the Company or its Subsidiaries (whether granted this Plan or otherwise), for a performance period ending in a calendar year, shall be made to the participant between January 1 and March 15 of the following calendar year; provided, however, that for purposes of determining compliance with Code section 409A, a payment will be considered to satisfy the requirement of this sentence if distribution is made no later than the end of the calendar year following the end of the applicable performance period.

SECTION 7

CHANGE IN CONTROL

7.1. <u>Change in Control</u>. Subject to the provisions of Section 3.2 and the authority of the Committee to take the actions permitted pursuant to Section 7.2, the occurrence of a Change in Control shall have the effect, if any, with respect to any Award as set forth in the Award Agreement or, to the extent not prohibited by the Plan or the Award Agreement, as provided by the Committee.

7.2. <u>Committee Actions On A Change in Control</u>. On a Change in Control, if the Plan is terminated by the Company or its successor without provision for the continuation of outstanding Awards hereunder, the Committee may cancel any outstanding Awards in return for cash payment of the current value of the Award, determined with the Award fully vested at the time of payment, provided that in the case of an Option, the amount of such payment will be the excess of value of the shares of Stock subject to the Option at the time of the transaction over the Exercise Price;

provided, further, that in the case of an Option, such Option will be cancelled with no payment if, as of the Change in Control, the value of the shares of Stock subject to the Option at the time of the transaction are equal to or less than the Exercise Price. However, in no event shall this Section 7.2 be construed to permit a payment if such payment would result in accelerated recognition of income or imposition of additional tax under Section 409A of the Code.

SECTION 8

COMMITTEE

8.1. <u>Administration</u>. The authority to control and manage the operation and administration of the Plan shall be vested in a committee (the "Committee") in accordance with this Section 8. The Committee shall be selected by the Board, and shall consist of two or more members of the Board. Unless otherwise provided by the Board, the Compensation Committee of the Board shall serve as the Committee. As a committee of the Board, the Committee is subject to the overview of the Board. If the Committee does not exist, or for any other reason determined by the Board, and to the extent not prohibited by Applicable Law, the Board may take any action under the Plan that would otherwise be the responsibility of the Committee.

8.2. <u>Selection of Committee</u>. So long as the Company is subject to Section 16 of the Exchange Act, the Committee shall be selected by the Board and shall consist of not fewer than two members of the Board or such greater number as may be required for compliance with Rule 16b-3 issued under the Exchange Act and shall be comprised of persons who are independent for purposes of applicable stock exchange listing requirements and who would meet the requirements of a "non-employee director" within the meaning of Rule 16b-3 under the Securities Exchange Act of 1934.

8.3. <u>Powers of Committee</u>. The Committee's administration of the Plan shall be subject to the following:

(a) Subject to the provisions of the Plan, the Committee will have the authority and discretion to select individuals who shall be Eligible Individuals and who, therefore, are eligible to receive Awards under the Plan. The Committee shall have the authority to determine the time or times of receipt of Awards, to determine the types of Awards and the number of shares of Stock covered by the Awards, to establish the terms, conditions, performance targets, restrictions, and other provisions of such Awards, to cancel or suspend Awards, and to accelerate the exercisability or vesting of any Award under circumstances designated by it. In making such Award determinations, the Committee may take into account the nature of services rendered by the respective employee, the individual's present and potential contribution to the Company's or a Related Company's success and such other factors as the Committee deems relevant.

- (b) To the extent that the Committee determines that the restrictions imposed by the Plan preclude the achievement of the material purposes of the Awards in jurisdictions outside the United States, the Committee will have the authority and discretion to modify those restrictions as the Committee determines to be necessary or appropriate to conform to applicable requirements or practices of jurisdictions outside of the United States.
- (c) The Committee will have the authority and discretion to interpret the Plan, to establish, amend, and rescind any rules and regulations relating to the Plan, to determine the terms and conditions of any Award Agreement made pursuant to the Plan, and to make all other determinations that may be necessary or advisable for the administration of the Plan.
- (d) Any interpretation of the Plan by the Committee and any decision made by it under the Plan is final and binding on all persons.
- (e) In controlling and managing the operation and administration of the Plan, the Committee shall take action in a manner that conforms to applicable corporate law.
- (f) Notwithstanding any other provision of the Plan, no benefit shall be distributed under the Plan to any person unless the Committee, in its sole discretion, determines that such person is entitled to benefits under the Plan.

8.4. <u>Delegation by Committee.</u> Except to the extent prohibited by Applicable Law, the Committee may allocate all or any portion of its responsibilities and powers to any one or more of its members and may delegate all or any part of its responsibilities and powers to any person or persons selected by it. Any such allocation or delegation may be revoked by the Committee at any time.

8.5. Information to be Furnished to Committee. The Company, Subsidiaries and any applicable Related Company shall furnish the Committee with such data and information as it determines may be required for it to discharge its duties. The records of the Company, Subsidiaries and any applicable Related Company as to an employee's or Participant's employment (or other provision of services), termination of employment (or cessation of the provision of services), leave of absence, reemployment

and compensation shall be conclusive on all persons unless determined to be incorrect. Participants and other persons entitled to benefits under the Plan must furnish the Committee such evidence, data or information as the Committee considers desirable to carry out the terms of the Plan.

8.6. Liability and Indemnification of Committee. No member or authorized delegate of the Committee shall be liable to any person for any action taken or omitted in connection with the administration of the Plan unless attributable to his own fraud or willful misconduct; nor shall the Company or any Related Company be liable to any person for any such action unless attributable to fraud or willful misconduct on the part of a director or employee of the Company or Related Company. The Committee, the individual members thereof, and persons acting as the authorized delegates of the Committee under the Plan, shall be indemnified by the Company against any and all liabilities, losses, costs and expenses (including legal fees and expenses) of whatsoever kind and nature which may be imposed on, incurred by or asserted against the Committee or its members or authorized delegates by reason of the performance of a Committee function if the Committee or its members or authorized delegates did not act dishonestly or in willful violation of the law or regulation under which such liability, loss, cost or expense arises. This indemnification shall not duplicate but may supplement any coverage available under any applicable insurance.

SECTION 9 AMENDMENT AND TERMINATION

The Board may, at any time, amend or terminate the Plan, and the Board or the Committee may amend any Award Agreement, provided that no amendment or termination may, in the absence of written consent to the change by the affected Participant (or, if the Participant is not then living, the affected beneficiary), adversely affect the rights of any Participant or beneficiary under any Award granted under the Plan prior to the date such amendment is adopted by the Board (or the Committee if applicable); and further provided that adjustments pursuant to Section 3.2 shall not be subject to the foregoing limitations of this Section 9; and further provided that the provisions of Section 4.7 (relating to Option repricing) cannot be amended unless the amendment is approved by the Company's stockholders. Approval by the Company's stockholders will be required for any material revision to the terms of the Plan, with the Committee's determination of "material revision" to take into account the exemptions under applicable stock exchange rules. No amendment or termination shall be adopted or effective if it would result in accelerated recognition of income or imposition of additional tax under Section 409A of the Code or, except as

otherwise provided in the amendment, would cause amounts that were not otherwise subject to Section 409A of the Code to become subject to Section 409A of the Code.

SECTION 10 GENERAL PROVISIONS

- 10.1. General Restrictions. Delivery of shares of Stock or other amounts under the Plan shall be subject to the following:
- (a) Notwithstanding any other provision of the Plan, the Company shall have no obligation to recognize an exercise of an Option or deliver any shares of Stock or make any other distribution of benefits under the Plan unless such exercise, delivery or distribution complies with all Applicable Laws (including, without limitation, the requirements of the United States Securities Act of 1933 and the securities laws of any other applicable jurisdiction), and the applicable requirements of any securities exchange or similar entity or other regulatory authority with respect to the issue of shares and securities by the Company.
- (b) To the extent that the Plan provides for issuance of share certificates to reflect the issuance of shares of Stock, the issuance may be effected on a non-certificated basis, to the extent not prohibited by Applicable Law, the By-laws of the Company.
- (c) To the extent provided by the Committee, any Award may be settled in cash rather than shares of Stock.

10.2. <u>Tax Withholding</u>. All distributions under the Plan are subject to withholding of all applicable taxes, and the Committee may condition the delivery of any shares of Stock or other benefits under the Plan on satisfaction of the applicable withholding obligations. Except as otherwise provided by the Committee and subject to Applicable Law, such withholding obligations may be satisfied (i) through cash payment by the Participant; (ii) through the surrender of shares of Stock which the Participant already owns; or (iii) through the surrender of shares of Stock to which the Participant is otherwise entitled under the Plan (including shares otherwise distributable pursuant to the Award); provided, however, that such shares of Stock under this clause (iii) may be used to satisfy not more than the maximum individual tax rate for the Participant in applicable jurisdiction for such Participant (based on the applicable rates of the relevant tax authorities (for example, federal, state, and local), including the Participant's share of payroll or similar taxes, as provided in tax law, regulations, or the authority's

administrative practices, not to exceed the highest statutory rate in that jurisdiction, even if that rate exceeds the highest rate that may be applicable to the specific Participant).

10.3. <u>Grant and Use of Awards</u>. In the discretion of the Committee, an Eligible Individual may be granted any Award permitted under the provisions of the Plan, and more than one Award may be granted to an Eligible Individual. Subject to Section 4.7 (relating to repricing), Awards may be granted as alternatives to or replacement of awards granted or outstanding under the Plan, or any other plan or arrangement of the Company or a Subsidiary or a Related Company (including a plan or arrangement of a business or entity, all or a portion of which is acquired by the Company or a Subsidiary or a Related Company). Subject to the overall limitation on the number of shares of Stock that may be delivered under the Plan, the Committee may use available shares of Stock as the form of payment for compensation, grants or rights earned or due under any other compensation plans or arrangements of the Company assumed in business combinations. Notwithstanding the provisions of Section 4.4, Options granted under the Plan in replacement for awards under plans and arrangements of the Company or a Subsidiary or a Related Company assumed in business combinations. Notwithstanding the provisions of Section 4.4, Options granted under the Plan in replacement for Exercise Prices that are less than the Fair Market Value of the shares of Stock at the time of the replacement grants, if the Committee determines that such Exercise Price is appropriate to preserve the economic benefit of the award. The provisions of this Section shall be subject to the provisions of Section 10.13.

10.4. <u>Dividends and Dividend Equivalents</u>. An Award (other than an Option) may provide the Participant with the right to receive dividend or dividend equivalent payments with respect to shares of Stock subject to the Award; provided, however, that no dividend or dividend equivalents granted in relation to Full Value Awards that are subject to vesting shall be settled prior to the date that such Full Value Award (or applicable portion thereof) becomes vested and is settled. Any such settlements, and any such crediting of dividends or dividend equivalents or reinvestment in shares of Stock, will be subject to the Company's By-laws as well as Applicable Law and further may be subject to such conditions, restrictions and contingencies as the Committee shall establish, including the reinvestment of such credited amounts in share of Stock equivalents. The provisions of this Section shall be subject to the provisions of Section 10.13.

10.5. Settlement of Awards. The obligation to make payments and distributions with respect to Awards may be satisfied through cash payments, the delivery of shares of Stock, the granting of replacement Awards, or combination thereof as the Committee shall determine. Satisfaction of any such obligations under an Award, which is sometimes referred to as "settlement" of the Award, may be subject to such conditions, restrictions and contingencies as the Committee shall determine. The Committee may permit or require the deferral of any Award payment or distribution, subject to such rules and procedures as it may establish, which may include provisions for the payment or crediting of interest or dividend equivalents, and may include converting such credits into deferred share of Stock equivalents. Except for Options designated at the time of grant or otherwise as intended to be subject to Section 409A of the Code, this Section 10.5 shall not be construed to permit the deferred settlement of Options, if such settlement would result in deferral of compensation under Treas. Reg. §1.409A-1(b)(5)(i)(A)(3) (except as permitted in Sections (i) and (ii) of that section). Each Subsidiary shall be liable for payment of cash due under the Plan with respect to any Participant to the extent that such benefits are attributable to the services rendered for that Subsidiary by the Participant. Any disputes relating to liability of a Subsidiary for cash payments shall be resolved by the Committee. The provisions of this Section shall be subject to the provisions of Section 10.13.

10.6. <u>Transferability</u>. Except as otherwise provided by the Committee, Awards under the Plan are not transferable except as designated by the Participant by will or by the laws of descent and distribution.

10.7. Form and Time of Elections. Unless otherwise specified herein, each election required or permitted to be made by any Participant or other person entitled to benefits under the Plan, and any permitted modification, or revocation thereof, shall be in writing filed with the Committee at such times, in such form, and subject to such restrictions and limitations, not inconsistent with the terms of the Plan, as the Committee shall require.

10.8. <u>Agreement With Company</u>. An Award under the Plan shall be subject to such terms and conditions, not inconsistent with the Plan, as the Committee shall, in its sole discretion, prescribe. The terms and conditions of any Award to any Participant shall be reflected in such form of written (including electronic) document as is determined by the Committee. A copy of such document shall be provided to the Participant, and the Committee may, but need not require that the Participant sign a copy

of such document. Such document is referred to in the Plan as an "Award Agreement" regardless of whether any Participant signature is required.

10.9. <u>Action by Company or Subsidiary</u>. Any action required or permitted to be taken by the Company or any Subsidiary or Related Company shall be by resolution of its board of directors, or by action of one or more members of the board (including a committee of the board) who are duly authorized to act for the board, or (except to the extent prohibited by Applicable Law or applicable rules of any stock exchange) by a duly authorized officer of such company.

10.10. <u>Gender and Number</u>. Where the context admits, words in any gender shall include any other gender, words in the singular shall include the plural and the plural shall include the singular.

10.11. Limitation of Implied Rights.

- (a) Neither a Participant nor any other person shall, by reason of participation in the Plan, acquire any right in or title to any assets, funds or property of the Company or any Subsidiary or Related Company whatsoever, including, without limitation, any specific funds, assets, or other property which the Company or any Subsidiary or Related Company, in its sole discretion, may set aside in anticipation of a liability under the Plan. A Participant shall have only a contractual right to the shares of Stock or amounts, if any, payable under the Plan, unsecured by any assets of the Company or any Subsidiary or Related Company, and nothing contained in the Plan shall constitute a guarantee that the assets of the Company or any Subsidiary or Related Company shall be sufficient to pay any benefits to any person.
- (b) The Plan does not constitute a contract of employment, and selection as a Participant will not give any participating employee or other individual the right to be retained in the employ of the Company or any Subsidiary or Related Company or the right to continue to provide services to the Company or any Subsidiary or Related Company, nor any right or claim to any benefit under the Plan, unless such right or claim has specifically accrued under the terms of the Plan. Except as otherwise provided in the Plan, no Award under the Plan shall confer upon the holder thereof any rights as a stockholder of the Company prior to the date on which the individual fulfills all conditions for receipt of such rights and is registered in the Company's Register of share of stockholders.

(c) All Stock and shares issued under any Award or otherwise are to be held subject to the provisions of the Company's By-laws and each Participant is deemed to agree to be bound by the terms of the Company's By-laws as they stand at the time of issue of any shares of Stock under the Plan.

10.12. Evidence. Evidence required of anyone under the Plan may be by certificate, affidavit, document or other information which the person acting on it considers pertinent and reliable, and signed, made or presented by the proper party or parties.

10.13. <u>Limitations under Section 409A</u>. The provisions of the Plan shall be subject to the following:

- (a) Awards will be designed and operated in such a manner that they are either exempt from the application of, or comply with, the requirements of Section 409A of the Code, except as otherwise determined in the sole discretion of the Administrator. The Plan and each Award Agreement under the Plan is intended to meet the requirements of Section 409A of the Code and will be construed and interpreted in accordance with such intent, except as otherwise determined in the sole discretion of the Administrator. To the extent that an Award or payment, or the settlement or deferral thereof, is subject to Section 409A of the Code the Award will be granted, paid, settled or deferred in a manner that will meet the requirements of Section 409A of the Code, such that the grant, payment, settlement or deferral will not be subject to the additional tax or interest applicable under Section 409A of the Code.
- (b) Neither Section 10.3 nor any other provision of the Plan shall be construed to permit the grant of an Option if such action would cause the Option being granted or the option or stock appreciation right being replaced to be subject to Section 409A of the Code, provided that this Section (b) shall not apply to any Option (or option or stock appreciation right granted under another plan) being replaced that, at the time it is granted or otherwise, is designated as being deferred compensation subject to Section 409A of the Code.
- (c) Except with respect to an Option that, at the time it is granted or otherwise, is designated as being deferred compensation subject to Section 409A of the Code, no Option shall condition the receipt of dividends with respect to an Option on the exercise of such Award, or otherwise provide for payment of such dividends in a

manner that would cause the payment to be treated as an offset to or reduction of the Exercise Price of the Option pursuant Treas. Reg. \$1.409A-1(b)(5)(i)(E).

(d) The Plan shall not be construed to permit a modification of an Award, or to permit the payment of a dividend or dividend equivalent, if such actions would result in accelerated recognition of taxable income or imposition of additional tax under Section 409A of the Code.

BRICKELL BIOTECH, INC.

EMPLOYEE STOCK PURCHASE PLAN

1. *Purpose of the Plan*. The purpose of this Brickell Biotech, Inc. Employee Stock Purchase Plan (the "Plan") is to provide the employees of Brickell Biotech, Inc. (the "Company") and its participating subsidiaries with a convenient means of purchasing shares of Company common stock from time to time at a discount to market prices through the use of payroll deductions. The Company intends that the Plan shall qualify as an "employee stock purchase plan" under Code § 423. Accordingly, the Plan will be construed so as to extend and limit Plan participation in any Offering subject to Code § 423 in a uniform and nondiscriminatory basis consistent with the requirements of Code § 423.

2. Definitions. The terms defined in this section are used (and capitalized) elsewhere in this Plan.

2.1. *"Affiliate"* means each domestic or foreign entity that is a "parent corporation" or "subsidiary corporation" of the Company, as defined in Code §§ 424(e) and 424(f) or any successor provisions.

2.2. "Board" means the Board of Directors of the Company.

2.3. "*Code*" means the Internal Revenue Code of 1986, as amended and in effect from time to time. For purposes of the Plan, references to sections of the Code shall be deemed to include any applicable regulations thereunder and any successor or similar statutory provisions.

2.4. "*Committee*" means the Compensation Committee of the Board (or such successor committee responsible for executive compensation matters).

2.5. "Common Stock" means the common stock, par value \$0.01 per share, of the Company.

2.6. "Company" means Brickell Biotech, Inc., a Delaware corporation, or any successor corporation.

2.7. "*Corporate Transaction*" means (i) a merger, consolidation or other reorganization of the Company with or into another corporation, or (ii) the sale of all or substantially all of the assets of the Company.

2.8. "Designated Affiliate" means any Affiliate which has been expressly designated by the Committee as a corporation whose Eligible Employees may participate in the Plan.

2.9. "*Eligible Compensation*" shall be defined from time to time by the Committee in its sole discretion with respect to any Offering and Purchase Period. Except as otherwise defined by the Committee from time to time in its sole discretion, (i) Eligible Compensation means the base salary

amount paid by the Company or any Designated Affiliate to a Participant in accordance with the Participant's terms of employment, (ii) Eligible Compensation includes contributions made by the Participant by payroll deduction to any qualified cash or deferred arrangement that forms part of a plan maintained by the Company or an Affiliate (while it is an Affiliate), or to a cafeteria plan maintained by the Company or an Affiliate (while it is an Affiliate), or to a cafeteria plan maintained by the Company or an Affiliate (while it is an Affiliate), or under any qualified transportation fringe benefit plan, and (iii) Eligible Compensation shall not include any commissions, overtime earnings, bonuses, employer contributions to a 401(k) or other retirement plan, amounts deferred to a non-qualified deferred compensation plan, any expense reimbursements or allowances, vacation pay in lieu of time off, coverage provided or amounts paid under any welfare benefit plan (unless provided above), amounts paid by an insurance company, amounts paid in a form other than cash and other fringe benefits, or any income (whether paid in Shares or cash) realized by the Participant as a result of participation in any equity-based compensation plan of the Company or an Affiliate.

2.10. *"Eligible Employee"* means any employee of the Company or a Designated Affiliate, except for any employee who, immediately after a right to purchase is granted under the Plan, would be deemed, for purposes of Code § 423(b)(3), to own stock possessing 5% or more of the total combined voting power or value of all classes of stock of the Company or any Affiliate. Notwithstanding the foregoing, with respect to any Offering, the Committee may provide for the exclusion of certain employees within the limitations described in Treasury Regulations §1.423-2(e)(1), (2) and (3).

2.11. "Enrollment Period" means the period of time prior to a Purchase Period during which Eligible Employees may elect to participate in the Plan as determined by the Committee for an Offering.

2.12. "Fair Market Value" of a Share of Common Stock as of any date means the closing sale price for a Share on the principal securities market on which the Shares trade on said date.

2.13. "Offering" means the right provided to Participants to purchase Shares under the Plan with respect to a Purchase Period.

2.14. "Offering Date" means the first Trading Day of a Purchase Period.

2.15. "*Participant*" means an Eligible Employee who has elected to participate in the Plan in the manner set forth in Section 4 and whose participation has not ended pursuant to Section 8.1 or Section 9.

2.16. "Plan" means this Brickell Biotech, Inc. Employee Stock Purchase Plan, as it may be amended from time to time.

2.17. "Purchase Date" means the last Trading Day of a Purchase Period.

2.18. "*Purchase Period*" means a period of time during which offers to purchase Common Stock are outstanding under the Plan. The Committee shall determine the length of each Purchase Period, which need not be uniform; provided that no Purchase Period shall exceed twenty-seven (27) months in length. A Purchase Period shall commence on such date as may be established by the Committee. Unless the Committee determines otherwise, the Purchase Period will be a period of six months beginning either (i) on January 1 of each calendar year and ending on the next June 30, or (ii) on July 1 in each calendar year and ending on the next December 31.

2.19. "*Recordkeeping Account*" means the account maintained in the books and records of the Company (or its agent) recording the amount contributed to the Plan by each Participant through payroll deductions.

2.20. "Shares" means shares of Common Stock.

2.21. "Trading Day" means a day on which the national stock exchanges in the United States are open for trading.

3. *Shares Available.* Subject to adjustment as provided in Section 14.1, the maximum number of Shares that may be sold by the Company to Eligible Employees under the Plan shall be 2,600,000 Shares. If the purchases by all Participants in an Offering would otherwise cause the aggregate number of Shares to be sold under the Plan to exceed the number specified in this Section 3, the Company shall make to each Participant in that Offering a pro rata allocation in a uniform and nondiscriminatory manner of the remaining number of Shares which may be sold under the Plan.

4. *Eligibility and Participation*. To be eligible to participate in the Plan for a given Purchase Period, an employee must be an Eligible Employee on the first day of such Purchase Period. An Eligible Employee may elect to participate in the Plan by filing an election form with the Company (or its agent) before the Offering Date for a Purchase Period that authorizes regular payroll deductions from Eligible Compensation beginning with the first payday in such Purchase Period and continuing until the Plan is terminated or the Eligible Employee withdraws from the Plan, modifies his or her authorization, or ceases to be an Eligible Employee, as hereinafter provided.

5. Amount of Common Stock Each Eligible Employee May Purchase.

5.1. *Purchase Amounts and Limitations*. Subject to the provisions of this Plan, each Participant shall be offered the right to purchase on the Purchase Date the maximum number of whole Shares that can be purchased with the balance in the Participant's Recordkeeping Account at the per Share price specified in Section 5.2. Notwithstanding the foregoing, no Participant shall be entitled to:

(a) the right to purchase Shares under this Plan and all other employee stock purchase plans (within the meaning of Code § 423(b)), if any, of the Company and its Affiliates that accrues at a rate which in the aggregate exceeds \$25,000 of Fair Market Value (determined on the Offering Date of a Purchase Period when the right is granted) for each calendar year in which such right is outstanding at any time; or

(b) purchase Shares in excess of 25,000 Shares per Offering (or such other maximum Share limit as established by the Committee in its sole discretion), with such limit subject to adjustment from time to time as provided in Section 14.1.

5.2. *Purchase Price*. Unless a different purchase price is established by the Committee for an Offering prior to the commencement of the applicable Purchase Period, the purchase price of each Share sold pursuant to this Plan will be the lesser of (i) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Period, or (ii) 85% of the Fair Market Value of such Share on the Purchase Date (such lesser price, the "Purchase Price"). In no event shall the Purchase Price be less than the lesser of (i) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Price be less than the lesser of (i) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Price be less than the lesser of (i) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Period, or (ii) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Period, or (ii) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Period, or (ii) 85% of the Fair Market Value of such Share on the Offering Date of the applicable Purchase Period, or (ii) 85% of the Fair Market Value of such Share on the Purchase Date.

6. Method of Participation.

6.1. *Notice and Date of Grant*. The Company shall give notice to each Eligible Employee of the opportunity to purchase Shares pursuant to this Plan and the terms and conditions of such Offering. The Company contemplates that for tax purposes, the Offering Date for a Purchase Period will be considered the date of the grant of the right to purchase such Shares.

6.2. *Contribution Elections*. Each Eligible Employee who desires to participate in the Plan for a Purchase Period shall signify his or her election to do so by completing an election with the Company (or its agent) in a manner approved by the Committee. An Eligible Employee may elect to have any whole percent of Eligible Compensation (that is, 1%, 2%, 3%, etc.) withheld as a payroll deduction, but not exceeding 10% per pay period (or such other maximum percentage as the Committee may establish from time to time prior to the commencement of an Offering). An election to participate in the Plan and to authorize payroll deductions as described herein must be made prior to the Offering Date of a Purchase Period in accordance with the rules set by the Committee for the Purchase Period, and shall be effective beginning with the first payday in the Purchase Period immediately following the filing of such election. Any election submitted shall remain in effect until the Plan is terminated or such Participant withdraws from the Plan, modifies his or her authorization, or ceases to be an Eligible Employee, as hereinafter provided.

6.3 *Additional Contributions*. If specifically provided by the Committee in connection with an Offering (including for purposes of complying with applicable local law), in addition to or instead of making contributions by payroll deductions, a Participant may make additional contributions to his or her Recordkeeping Account through the payment by cash or check prior to a Purchase Date. A Participant may make such additional contributions into his or her Recordkeeping Account only if the Participant has not already had the maximum permitted amount withheld during the Offering through payroll deductions, subject to the limitations set forth in Section 5.1.

6.4. *Offering Terms and Conditions*. Each Offering shall consist of a single Purchase Period and shall be in such form and shall contain such terms and conditions as the Committee shall deem

appropriate, consistent with the terms of the Plan. The Committee may provide for separate Offerings for different Designated Affiliates, and the terms and conditions of the separate Offerings, including the applicable Purchase Period, need not be consistent. Any Offering shall comply with the requirement of Code § 423 that all Participants shall have the same rights and privileges for such Offering. The terms and conditions of any Offering shall be incorporated by reference into the Plan and treated as part of the Plan.

7. Recordkeeping Accounts.

7.1. *Crediting Payroll Deduction Contributions*. The Company (or its agent) shall maintain a Recordkeeping Account for each Participant. Payroll deductions pursuant to Section 6 will be credited to such Recordkeeping Accounts on or within a reasonable amount of time following each payday.

7.2. No Interest Payable. No interest will be credited to a Participant's Recordkeeping Account (unless required under local law).

7.3. *No Segregation of Accounts*. The Recordkeeping Account is established solely for accounting purposes, and all amounts credited to the Recordkeeping Account will remain part of the general assets of the Company and need not be segregated from other corporate funds (unless required under local law).

7.4. *Additional Contributions*. A Participant may not make any separate cash payment into a Recordkeeping Account, except as may be permitted in accordance with Section 6.3, and any such additional contributions will be credited to the Recordkeeping Accounts within a reasonable amount of time following receipt by the Company.

8. Right to Adjust Participation; Withdrawals from Recordkeeping Account.

8.1. *Withdrawal from Plan.* A Participant may at any time withdraw from the Plan by complying with the rules set by the Committee. If a Participant withdraws from the Plan, the Company will pay to the Participant in cash the entire balance in such Participant's Recordkeeping Account and no further deductions will be made from the Participant's Eligible Compensation during such Purchase Period. A Participant who withdraws from the Plan will not be eligible to reenter the Plan until the next succeeding Purchase Period, and any such reentry shall be through the enrollment process described in Section 6.2.

8.2. Adjusting Level of Participation. A Participant may adjust his or her rate of payroll deduction contributions to the Plan as follows:

(a) A Participant may, by written notice during an Enrollment Period, direct the Company to increase or decrease his or her rate of payroll deduction contributions, with such change to be effective as of the first day of the next Purchase Period.

(b) A Participant may, by written notice that complies with the rules set by the Committee, direct the Company to decrease his or her rate of payroll deduction contributions during a Purchase Period to 0%, which shall be considered a suspension of contributions and shall become effective as soon as reasonably practicable. Any Participant who has decreased his or her rate of payroll deductions to 0% and does not increase such rate of payroll deductions from 0% to at least 1% in accordance with Section 8.2(a) during the next Enrollment Period will be withdrawn from the Plan effective as of the first day of that next Purchase Period.

8.3. *Submission of Notices*. Notification of a Participant's election to withdraw from the Plan as provided in Section 8.1 or to change his or her rate of payroll deductions as provided in Section 8.2 shall be made by completing an updated election or notice with the Company (or its agent) in a manner approved by the Committee. The Committee may promulgate rules regarding the time and manner for submitting any such updated election or notice, which may include a requirement that the election or notice be on file for a reasonable period before it will be effective.

8.4. *Adjustments by the Company*. To the extent necessary to comply with Code § 423(b)(8) or Section 5.1, a Participant's payroll deduction contributions to the Plan may be decreased by the Company to 0% at any time during a Purchase Period.

9. Termination of Employment.

9.1. *Refund of Recordkeeping Account.* If the employment of a Participant is terminated for any reason, including death, disability, or retirement, the entire balance in the Participant's Recordkeeping Account will be refunded in cash to the Participant within 30 days after the date of termination of employment. For purposes of the Plan, a Participant will not be deemed to have terminated employment while the Participant is on sick leave, military leave or other leave of absence approved by the Company. Where the period of leave exceeds 90 days and the Participant's right to reemployment is not guaranteed either by statute or by contract, the employment relationship shall be deemed to have terminated on the ninety-first day of such leave. Unless determined otherwise by the Committee in a manner that is permitted by, and in compliance with Code § 423, a Participant whose employment transfers between entities through a termination with an immediate rehire (with no break in service) by the Company or a Designated Affiliate shall not be treated as a termination under the Plan.

9.2. Designation of Beneficiary. If permitted by the Committee, a Participant may file a beneficiary designation for who is to receive the Participant's Recordkeeping Account or Share subaccount, if any, following the death of a Participant. If no beneficiary is named, the beneficiary shall be the Participant's spouse, or if none, the Participant's estate. All beneficiary designations will be in such form and manner as the Committee may designate from time to time.

10. Purchase of Shares.

10.1. *Number of Shares Purchased*. As of each Purchase Date, the balance in each Participant's Recordkeeping Account will be used to purchase the maximum number of whole Shares (subject to the limitations of Section 5.1) at the Purchase Price determined in accordance with Section 5.2, unless the Participant has filed an appropriate form with the Company in advance of that date to withdraw from the Plan in accordance with Section 8.1. Any amount remaining in a Participant's Recordkeeping Account that represents the Purchase Price for any fractional share will be carried over in the Participant's Recordkeeping Account to the next Purchase Period. Any amount remaining in a Participant's Recordkeeping Account to the purchase by reason of the limitations of Section 5.1 or under the circumstances described in Section 3 will be refunded to the Participant.

10.2. *Conversion of Foreign Currency*. In circumstances where payroll deductions have been taken from a Participant's Eligible Compensation in a currency other than United States dollars, Shares shall be purchased by converting the balance in the Participant's Recordkeeping Account to United States dollars at the exchange rate in effect for payroll purposes for the month in which the Purchase Date occurs as determined by the Company's finance department or at such other exchange rate determined by the Committee or its delegate for this purpose, and such dollar amount shall be used to purchase Shares as of the Purchase Date.

10.3. Crediting of Shares. Promptly after the end of each Purchase Period, the number of Shares purchased by all Participants as of the applicable Purchase Date shall be issued and delivered to an agent selected by the Company. Delivery of the shares to the agent shall be effected by an appropriate book-entry in the stock register maintained by the Company's transfer agent or delivery of a certificate. The agent will hold the Shares for the benefit of all Participants who have purchased Shares and will maintain a Share subaccount for each Participant reflecting the number of Shares credited to each Participant. Each Participant will be entitled to direct the voting by the agent of all Shares credited to such Participant's Share subaccount, and the agent may reinvest any dividends paid on Shares credited to a Participant's Share subaccount in additional Shares in accordance with such rules as the Committee may prescribe. Each Participant may also direct the agent to sell any or all of the Shares credited to the Participant's Share subaccount and distribute the net proceeds of such sale to the Participant.

10.4 *Withdrawal of Shares from Share Subaccount*. Except for sales through the agent as provided in Section 10.3, a Participant may not withdraw Shares or otherwise transfer Shares from the Participant's Share subaccount.

11. *Rights as a Shareholder*. A Participant shall not be entitled to any of the rights or privileges of a shareholder of the Company with respect to Shares offered for purchase under the Plan, including the right to vote or direct the voting or to receive any dividends that may be declared by the Company, until (i) the Participant actually has paid the Purchase Price for such Shares and (ii) such Shares have been issued and delivered, as provided in Section 10.3.

12. *Rights Not Transferable*. A Participant's rights under this Plan are exercisable only by the Participant during his or her lifetime, and may not be sold, pledged, assigned, transferred or disposed of in any manner other than by will or the laws of descent and distribution. Any attempt to sell, pledge, assign, transfer or dispose of the same shall be void and without effect. The amounts credited to a Recordkeeping Account may not be sold, pledged, assigned, transferred or disposed of in any way, and any attempted sale, pledge, assignment, transfer or other disposition of such amounts will be void and without effect.

13. Administration of the Plan.

13.1. *Authority of the Committee*. This Plan shall be administered by the Committee. Subject to the express provisions of the Plan and applicable law, and in addition to other express powers and authorizations conferred on the Committee by the Plan, the Committee shall have full power and authority to:

(a) Determine when each Purchase Period under this Plan shall occur, and the terms and conditions of each related Offering (which need not be identical);

(b) Designate from time to time which Affiliates of the Company shall be eligible to participate in the Plan;

(c) Construe and interpret the Plan and establish, amend and revoke rules, regulations and procedures for the administration of the Plan. The Committee may, in the exercise of this power, correct any defect, omission or inconsistency in the Plan, in such manner and to the extent it may deem necessary, desirable or appropriate to make the Plan fully effective;

(d) Exercise such powers and perform such acts as the Committee may deem necessary, desirable or appropriate to promote the best interests of the Company and its Designated Affiliates and to carry out the intent that the Offerings made under the Plan are treated as qualifying under Code § 423(b);

(e) As more fully described in Section 18, to adopt such rules, procedures and sub-plans as may be necessary, desirable or appropriate to permit participation in the Plan by employees who are foreign nationals or employed outside the United States by a non-U.S. Designated Affiliate, and to achieve tax, securities law and other compliance objectives in particular locations outside the United States; and

(f) Adopt and amend, as the Committee deems appropriate, a Plan rule specifying that Shares purchased by a Participant during a Purchase Period may not be sold by the Participant for a specified period of time after the Purchase Date on which the Shares were purchased by the Participant, and establish such procedures as the Committee may deem necessary to implement such rule.

13.2. Interpretations and Decisions by the Committee. Unless otherwise expressly provided in the Plan, all designations, determinations, interpretations, and other decisions under or with respect to the Plan shall be within the sole discretion of the Committee, may be made at any time and shall be final, conclusive, and binding upon all persons, including the Company, any Affiliate, any Participant and any Eligible Employee.

13.3. Delegation by the Committee. Subject to the terms of the Plan and applicable law, the Committee may delegate ministerial duties associated with the administration of the Plan to such of the Company's officers, employees or agents as the Committee may determine.

13.4. *Indemnification*. No member of the Board or Committee shall be liable for any action taken or determination made in good faith with respect to the Plan. In addition to such other rights of indemnification as they may have as members of the Board or officers or employees of the Company or a Designated Affiliate, members of the Board and Committee and any officers or employees of the Company or Designated Affiliate to whom authority to act for the Committee is delegated shall be indemnified by the Company from and against any and all liabilities, costs and expenses incurred by such persons as a result of any act or omission to act in connection with the performance of such person's duties, responsibilities and obligations under the Plan if such person has acted in good faith and in a manner that he or she reasonably believes to be in, or not opposed to, the best interests of the Company.

14. Changes in Capitalization and Corporate Transactions.

14.1. *Adjustments*. In the event of any change in the Common Stock of the Company by reason of a stock dividend, stock split, reverse stock split, corporate separation, recapitalization, merger, consolidation, combination, exchange of shares and the like, the Committee shall make such equitable adjustments as it deems appropriate in the aggregate number and class of Shares or other securities available under this Plan, the Share limitation referred to in Section 5.1(b) of the Plan, and the number, class and Purchase Price of Shares or other securities subject to purchase under any pending Offering.

14.2. Corporate Transactions. In the event of a Corporate Transaction, each right to acquire Shares on any Purchase Date that is scheduled to occur after the date of the consummation of the Corporate Transaction may be continued or assumed or an equivalent right may be substituted by the surviving or successor corporation or a parent or subsidiary of such corporation. If such surviving or successor corporation or parent or subsidiary thereof refuses to continue, assume or substitute for such outstanding rights, then the Board may, in its discretion, either terminate the Plan or shorten the Purchase Period then in progress by setting a new Purchase Date for a specified date before the date of the consummation of the Corporate Transaction. Each Participant shall be notified in writing, prior to any new Purchase Date, that the Purchase Date for the existing Offering has been changed to the new Purchase Date and that the Participant's right to acquire Shares will be exercised automatically on the new Purchase Date unless prior to such date the Participant's employment has been terminated



or the Participant has withdrawn from the Plan. In the event of a dissolution or liquidation of the Company, any Offering and Purchase Period then in progress will terminate immediately prior to the consummation of such action, unless otherwise provided by the Board.

15. *Amendment or Suspension of Plan*. The Committee, in its sole discretion, may at any time suspend this Plan or amend it in any respect, but no such amendment may, without shareholder approval, increase the number of shares reserved under this Plan, or effect any other change in the Plan that would require shareholder approval under applicable law or regulations or the rules of any securities exchange on which the Shares may then be listed, or to maintain compliance with Code § 423. No such amendment or suspension shall adversely affect the rights of Participants pursuant to Shares previously acquired under the Plan. During any suspension of the Plan, no new Offering or Purchase Period shall begin and no Eligible Employee shall be offered any new right to purchase Shares under the Plan or any opportunity to elect to participate in the Plan, and any existing payroll deduction authorizations shall be suspended, but any such right to purchase Shares previously granted for a Purchase Period that began prior to the Plan suspension shall remain subject to the other provisions of this Plan and the discretion of the Board and the Committee with respect thereto.

16. *Effective Date and Term of Plan*. The Plan will become effective on the date it is approved by the shareholders of the Company, which approval must be within 12 months of the date the Plan is adopted by the Board. The Plan and all rights of Participants hereunder shall terminate (i) at any time, at the discretion of the Committee, or (ii) upon the completion of any Offering under which the limitation on the total number of Shares to be issued during the entire term of the Plan, as determined in accordance with Section 3, has been reached. Except as otherwise determined by the Board, upon termination of this Plan, the Company shall pay to each Participant cash in an amount equal to the entire remaining balance in such Participant's Recordkeeping Account.

17. *Governmental Regulations and Listing*. All rights granted or to be granted to Eligible Employees under this Plan are expressly subject to all applicable laws and regulations and to the approval of all governmental authorities required in connection with the authorization, issuance, sale or transfer of the Shares reserved for this Plan, including, without limitation, there being a current registration statement of the Company under the Securities Act of 1933, as amended, covering the Shares purchasable on the Purchase Date applicable to such Shares. If applicable, all such rights hereunder are also similarly subject to effectiveness of an appropriate listing application to a national securities exchange covering the Shares issuable under the Plan upon official notice of issuance.

18. *Rules for Foreign Jurisdictions*. The Committee may adopt rules, procedures or subplans relating to the operation and administration of the Plan to accommodate the specific requirements of local laws and procedures. Without limiting the generality of the foregoing, the Committee is specifically authorized to adopt rules and procedures regarding handling of payroll deductions, payment of interest, conversion of local currency, payroll tax, the definition of Eligible Compensation, withholding procedures and handling of stock certificates that vary with local requirements.

19. Miscellaneous.

19.1. *Effect on Employment Status*. This Plan shall not be deemed to constitute a contract of employment between the Company or any Designated Affiliate and any Participant, nor shall it interfere with the right of the Company (or any Affiliate) to terminate the employment of any Participant and treat him or her without regard to the effect that such treatment might have upon him or her under this Plan.

19.2. *Governing Law*. This Plan, and all agreements hereunder, shall be construed in accordance with and governed by the laws of the State of Delaware.

19.3. *Electronic Documentation and Signatures*. Any reference in the Plan to election or enrollment forms, notices, authorizations or any other document to be provided in writing shall include the provision of any such form, notice, authorization or document by electronic means, including through the Company's intranet or with the Company's agent, and any reference in the Plan to the signing of any document shall include the authentication of any such document provided in electronic form, in each case in accordance with procedures established by the Committee.

19.4. Book-Entry and Electronic Transfer of Shares. Any reference in this Plan to the issuance or transfer of a stock certificate evidencing Shares shall be deemed to include, in the Committee's discretion, the issuance or transfer of such Shares in book-entry or electronic form. Uncertificated Shares shall be deemed delivered for all purposes of this Plan when the Company or its agent shall have provided to the recipient of the Shares a notice of issuance or transfer by electronic mail (with proof of receipt) or by United States mail, and have recorded the issuance or transfer in its records.

19.5. Registration of Share Accounts and Certificates. Any Share account contemplated by Section 10.3 and certificate to be issued to a Participant shall be registered in the name of the Participant, or jointly in the name of the Participant and another person, as the Participant may direct on an appropriate form filed with the Company or the agent.

19.6. *Code § 409A*. The Plan is exempt from the application of Code § 409A and any ambiguities herein will be interpreted to so be exempt from Code § 409A. In furtherance of the foregoing and notwithstanding any provision in the Plan to the contrary, if the Committee determines that an option granted under the Plan may be subject to Code § 409A or that any provision in the Plan would cause an option under the Plan to be subject to Code § 409A, the Committee may amend the terms of the Plan and/or of an outstanding Offering under the Plan, or take such other action as the Committee determines is necessary or appropriate, in each case, without the Participant's consent, to exempt any outstanding option or future option that may be granted under the Plan from or to allow any such options to comply with Code § 409A, but only to the extent any such amendments or actions by the Committee would not violate Code § 409A. Notwithstanding the foregoing, the Company and the Committee shall have no liability to a Participant or any other party if the option to purchase Shares under the Plan that is intended to be exempt from or compliant with Code § 409A is not exempt or compliant or for any action taken by the Committee with respect thereto. The Company



makes no representations that the option to purchase Shares under the Plan is compliant with Code § 409A.

19.7. *Severability*. If any provision of the Plan is or becomes or is deemed to be invalid, illegal, or unenforceable for any reason in any jurisdiction or as to any Participant, such invalidity, illegality or unenforceability shall not affect the remaining parts of the Plan and the Plan shall be construed and enforced as to such jurisdiction or Participant as if the invalid, illegal or unenforceable provision had not been included.